Author's response to reviews

Title: Early life adversity and C-reactive protein in diverse populations of older adults: a cross-sectional analysis from the International Mobility in Aging Study (IMIAS)

Authors:

Annie Li (annie.li@mail.mcgill.ca)
Mai Thanh Tu (mai.thanh.tu@umontreal.ca)
Ana Carolina Sousa (acapas@gmail.com)
Beatriz Alvarado (alvaradb@queensu.ca)
Georges Karna Kone (karnageorges@gmail.com)
Jack Guralnik (jguralnik@epi.umaryland.edu)
Maria Victoria Zunzunegui (maria.victoria.zunzunegui@umontreal.ca)

Version: 2 Date: 4 March 2015

Author's response to reviews:

There are no revisions in this version of the manuscript.